Author's response to reviews

Title: A Single-Blind Randomized Controlled Trial to evaluate the effect of extended counseling on uptake of pre-Antiretroviral care in Eastern Uganda.

Authors:

Lubega Muhamadi Dr (dlmuhamadi@yahoo.co.uk)
Nazarius M Tumwesigye Dr (naz@musph.ac.ug)
Daniel Kadobera (dkadobera@yahoo.com)
Gaetano Maronne (Gaetano.Marrone@ki.se)
Fred Wabwire-mangeni (fwabwire@musph.ac.ug)
George Pariyo (gpariyo@musph.ac.ug)
Stefan Peterson (Stefan.peterson@ki.se)
AnnaMia Ekström (anna.mia.ekstrom@ki.se)

Version: 7 Date: 1 July 2011

Author's response to reviews: see over
Dear Sir/Madam

Re: A Single blind Randomized Controlled Trial to evaluate the effect of extended counseling on uptake of Pre-antiretroviral care in Eastern Uganda (MS: 1581605345514791)

Lubega Muhamadi, Nazarius M Tumwesigye, Daniel Kadobera, Gaetano Maronne, Fred Wabwire-Mangen, George Pariyo, Stefan Peterson and Anna-Mia Ekström

Please find our point by point response to the advice as offered by the editors concerning the above captioned article.

1. A conclusions paragraph has now been separately inserted explaining the relevance and importance of the article.
2. A list of abbreviations has also now been inserted in the section before the references
3. All figures have now been uploaded as separate files as advised
4. All the tables have now been inserted after the reference list as advised by the editors

On behalf of all the authors please once again accept my sincere gratitude for your recommendations and advice aimed at improving our manuscript.

Sincerely yours

Dr Lubega Muhamadi
Corresponding author